\begin{table}%t1
\caption{\label{catalog}\integral\ AGN catalogue.  The column {\it Method} indicates the method used to determine the black hole mass, see Sect.~\ref{bhmasses} for details.}
\small%\centering
\par
\begin{tabular}{lccrrrrcc}
\hline\hline
Name &  Type & $z$ &  RA &  Dec & ISGRI &  JEM-X & $\log M_{\rm BH}$ & Method\\
 &   &  & $[\rm deg]$ & $[\rm deg]$ & $[\rm ks]$ &  $[\rm ks]$ & $[M_\odot]$\\ 
\hline
IGR J00040+7020$^x$ & Sy2   & 0.096  &  1.00638  &  70.32125 & 1947.3 & 20.9\\
IGR J00254+6822 & Sy2   & 0.012  &  6.38092  &  68.36147 & 2534.6 & 178.9\\
IGR J00335+6126 & Sy1   & 0.105  &  8.3265   &  61.46178 & 3338.7 &
356.5 & $8.5 \pm 0.5^a$ & LL/CL\\
1ES 0033+59.5   & BLLac & 0.086  &  8.96929  &  59.83461 & 3338.7 & 356.4\\
Mrk 348         & Sy2   & 0.0151 & 12.19642  &  31.95697 &  150.7 &
21.1 & $7.2 \pm 0.7^b$ & SO\\
NGC 418$^x$         & AGN   & 0.0190 & 17.64842  &--30.22128 &   38.5 &  --\\
NGC 526A        & Sy1.5 & 0.0191 & 20.97583  &--35.06528 &   73.7 &
-- & $8.1 \pm 0.7^c$ & S\\
ESO 297-18      & Sy2   & 0.0252 & 24.65492  &--40.01131 &   59.5 &
-- & $9.7 \pm 0.5^n$ & KM\\
IGR J01528-0326 & Sy2   & 0.0167 & 28.20375  & --3.44749 &  837.1 &  79.6\\
NGC 788         & Sy2   & 0.0136 & 30.27687  & --6.81553 &  926.6 &
117.9 & $7.5 \pm 0.7^b$ & SO\\
Mrk 590         & Sy1.2 & 0.0264 & 33.63984  & --0.76669 &  936.0 &
266.6 & $7.14 {+0.1 \atop -0.09}^d$ & R\\
IGR J02097+5222 & Sy1   & 0.0492 & 32.40700  &  52.44543 &  698.4 &   2.0\\
SWIFT J0216.3+5128 & Sy2 &0.0288 & 34.11292  &  51.42375 &  481.9 &   2.0\\
Mrk 1040        & Sy1.5 & 0.0167 & 37.06079  &  31.31094 &   42.8 &
-- & $7.6 \pm 0.3^e$ & S\\
IGR J02343+3229 & Sy2   & 0.0162 & 38.57500  &  32.48333 &   70.5 &  --\\
NGC 985         & Sy1   & 0.0431 & 38.65738  & --8.78761 &  808.1 &
48.3 & $8.9 \pm 0.5^n$ & KM\\
NGC 1052        & Sy2   & 0.0050 & 40.27000  & --8.25578 &  707.0 &
31.1 & $8.2 \pm 0.3^e$ & S\\
RBS 345         & Sy1   & 0.0690 & 40.56667  &   5.53000 &  443.2 &  --  \\
NGC 1068        & Sy2   & 0.0288 & 40.67012  & --0.01344 &  914.4 &
44.8 & $7.2 \pm 0.1^e$ & M\\
QSO B0241+62    & Sy1   & 0.0446 & 41.24042  &  62.46847 &  711.9 &  35.2\\
IGR J02466-4222 & AGN   & 0.0695 & 41.65375  &--42.36600 &  191.4 &  21.9\\
IGR J02501+5440 & Sy2   & 0.015  & 42.67417  &  54.70419 &  773.8 &  13.5\\
MCG-02-08-014   & Sy2   & 0.0168 & 43.09750  & --8.51042 &  528.1 &   4.7\\
NGC 1142        & Sy2   & 0.0288 & 43.80133  & --0.18381 &  634.1 &
-- & $9.4 \pm 0.5^n$ & KM\\
QSO B0309+411   & Sy1   & 0.136  & 48.25817  &  41.33366 &  488.4 & 303.1\\
IGR J03184-0014$^x$&QSO & --     & 49.60000  & --0.22889 &  --    &  --\\
NGC 1275        & Sy2   & 0.0176 & 49.95067  &  41.51170 &  506.8 &
282.0 & $8.5 \pm 0.7^e$ & S\\
1H 0323+342     & Sy1   & 0.0629 & 51.17150  &  34.17941 &  431.6 &   0.9\\
IGR J03334+3718 & Sy1.5 & 0.0547 & 53.32833  &  37.30305 &  463.4 &   5.6\\
NGC 1365        & Sy1.5 & 0.0055 & 53.40208  &--36.13806 &  143.3 &
3.0 & $7.7 \pm 0.3^f$ & S\\
IGR J03532-6829 & BLLac & 0.0870 & 58.30833  &--68.48306 &  740.8 &   9.4\\
3C 111          & Sy1   & 0.0485 & 64.58867  &  38.02661 &  160.1 &
-- & $9.6 \pm 0.8^c$ & B\\
3C 120          & Sy1   & 0.0330 & 68.29623  &   5.35434 &  365.9 &
79.3 & $7.7 \pm 0.2^g$ & R\\
UGC 3142        & Sy1   & 0.0217 & 70.94537  &  28.97194 &  449.6 &  39.9\\
LEDA 168563     & Sy1   & 0.0290 & 73.01958  &  49.54583 &  112.5 & --\\
ESO 33-2        & Sy2   & 0.0181 & 73.99834  &--75.54056 & 1110.4 & --\\
4U 0517+17      & Sy1.5 & 0.0179 & 77.68958  &  16.49861 & 1145.4 &  19.8\\
Ark 120         & Sy1   & 0.0327 & 79.04784  & --0.15017 &  571.9 &
63.1 & $8.18 {+0.05 \atop -0.06}^g$ & R\\
IGR J05270-6631$^x$& QSO   & 0.978  & 81.56021  &--66.5125  &  631.7 &
323.4 & $8.4 \pm 0.5^a$ & LL/CL\\
PKS 0528+134$^x$& blazar& 2.060  & 82.73507  &  13.53199 &  917.6 & 157.3\\
NGC 2110        & Sy2   & 0.0078 & 88.04742  & --7.45622 &   20.5 & --
& $8.3 \pm 0.3^e$ & S\\
MCG+08-11-011   & Sy1.5 & 0.0205 & 88.72338  &  46.43934 &   50.8 & --
& $8.1 \pm 0.6^c$ & SO\\
IRAS 05589+2828 & Sy1   & 0.0330 & 90.54042  &  28.47139 & 1689.9 &  29.0 \\
SWIFT J0601.9-8636 & Sy2 &0.0064 & 91.41292  &--86.63111 &  177.8 &
31.9 & $7.9 \pm 0.5^n$ & KM\\
IGR J06117-6625 & Sy1.5 & 0.230  & 92.95208  &--66.40847 & 1122.2 &  97.2\\
Mrk 3           & Sy2   & 0.0135 & 93.90129  &  71.03748 &  814.9 &
55.4 & $8.7 \pm 0.3^e$ & S\\
IGR J06239-6052 & Sy2   & 0.0405 & 95.94004  &--60.97927 &  598.0 & --\\
IGR J06292+4858$^x$ & BLLac & 0.097  & 97.300    &  48.97389 &  --    & --\\
PKS 0637-752    & Sy1   & 0.651  & 98.94379  &--75.27133 & 1066.6 &
22.7 & $9.4 \pm 0.5^e$ & CL\\
Mrk 6           & Sy1.5 & 0.0188 & 103.0513  &  74.42689 &  819.4 &
109.7 & $8.2 \pm 0.5^n$ & KM\\
QSO B0716+714   & BLLac & 0.3    & 110.4727  &  71.34343 &  863.8 & 162.8\\
LEDA 96373      & Sy2   & 0.0294 & 111.6096  &--35.90583 &  180.9 &   5.6\\
IGR J07437-5137 & Sy2   & 0.025  & 115.9208  &--51.61694 &  947.3 &   --\\
IGR J07565-4139 & Sy2   & 0.021  & 119.0817  &--41.62836 & 1703.0 &  50.6\\
IGR J07597-3842 & Sy1   & 0.040  & 119.9242  &--38.73223 & 1288.1 &  31.8 
& $8.3 \pm 0.5^p$ & LL \\
ESO 209-12      & Sy1.5 & 0.0405 & 120.4900  &--49.77833 & 2288.6 & 117.7\\
PG 0804+761     & Sy1   & 0.10   & 122.7444  &  76.04514 &  694.9 &
44.9 & $8.84 {+0.05 \atop -0.06}^g$ & R\\
Fairall 1146    & Sy1.5 & 0.0316 & 129.6279  &--35.99306 & 1689.9 &  26.2\\
QSO B0836+710   & BLLac & 2.1720 & 130.3515  &  70.89506 &  754.4 &  29.7\\
IGR J09026-4812$^m$ & Sy1&0.039  & 135.6555  &--48.22608 & 3773.3 & 470.1\\
SWIFT J0917.2-6221& Sy1 & 0.0573 & 139.0392  &--62.32486 &  669.0 &  16.1\\
IGR J09253+6929 & Sy1   & 0.039  & 141.321   &  69.488   &  324.1 & --
& $7.6 \pm 0.5^a$ & LL/CL\\
Mrk 110         & NLS1  & 0.0353 & 141.3036  &  52.28625 &   54.2 & --
& $7.42 {+0.09 \atop -0.1}^g$ & R\\
IGR J09446-2636 & Sy1.5 & 0.1425 & 146.1500  &--26.60000 &  157.1 & --\\
NGC 2992       & Sy1   & 0.0077 & 146.4252  &--14.32639 &  383.3 &
66.7 & $7.7 \pm 0.3^e$ & S\\
\hline
MCG-05-23-016   & Sy2   & 0.0085 & 146.9173  &--30.94886 &  130.2 & --
& $6.3 \pm 0.5^c$ & SO\\
IGR J09523-6231 & Sy1.5 & 0.252  & 148.0854  &--62.54333 & 1059.1 & 111.1\\
NGC 3081        & Sy2   & 0.0080 & 149.8731  &--22.82628 &  255.3 & --
& $7.4 \pm 0.3^b$ & S\\
SWIFT J1009.3-4250& Sy2 & 0.033  & 152.4512  &--42.81222 &  437.4 & --\\
IGR J10147-6354 & Sy1.2 & 0.202  & 153.6750  &--63.89194 & 1182.9 &
104.6 & $8.6 \pm 0.5^a$ & LL/CL\\
NGC 3227        & Sy1.5 & 0.0039 & 155.8776  &  19.86492 &  142.0 & --
& $7.3 {+0.2 \atop -0.1}^h$ & K\\
NGC 3281        & Sy2   & 0.0107 & 157.9669  &--34.85369 &  211.3 &
41.9 & $8.0 \pm 0.5^n$ & KM\\
SWIFT J1038.8-4942&Sy1.5& 0.060  & 159.6875  &--49.78194 &  908.6 & 10.2\\
IGR J10404-4625 & Sy2   & 0.0237 & 160.0928  &--46.42353 &  673.6 & --\\
Mrk 421         & BLLac & 0.0300 & 166.1138  &  38.20883 &  826.1 &
482.4 & $8.3 \pm 0.3^e$ & S\\
IGR J11366-6002 & Sy2   & 0.014  & 174.1754  &--60.05217 & 2196.0 & 315.7 \\
NGC 3783        & Sy1   & 0.0097 & 174.7574  &--37.73853 &   22.9 & --
& $7.47 {+0.07 \atop -0.09}^g$ & R\\
IGR J12026-5349 & Sy2   & 0.028  & 180.6985  &--53.83547 & 1535.9 & 116.6\\
NGC 4051        & Sy1.5 & 0.0023 & 180.7901  &  44.53144 &  793.2 & --
& $6.3 \pm 0.2^g$ & R\\
NGC 4138        & Sy1.5 & 0.0030 & 182.3745  &  43.68500 &  798.4 &
31.4 & $6.8 \pm 0.5^n$ & KM\\
NGC 4151        & Sy1.5 & 0.0033 & 182.6364  &  39.40545 &  820.3 &
548.3 & $7.5 {+0.1 \atop -0.6}^h$ & K\\
NGC 4180        & AGN   & 0.0070 & 183.2627  &   7.03881 & 1016.8 &  72.1\\
Was 49          & Sy2   & 0.0610 & 183.5742  &  29.52872 &  931.4 & --\\
Mrk 766         & Sy1.5 & 0.0129 & 184.6110  &  29.81267 &  994.6 & --
& $6.5 \pm 0.3^i$ & S\\
NGC 4258        & Sy1.5 & 0.0015 & 184.7397  &  47.30397 &  816.2 &
4.6 & $7.59 \pm 0.01^i$ & M\\
4C 04.42        & BLLac & 0.9650 & 185.5940  &   4.22106 & 1277.5 & 232.6\\
Mrk 50          & Sy1   & 0.0234 & 185.8506  &   2.67911 & 1320.8 & 264.8\\
NGC 4388        & Sy2   & 0.0084 & 186.4455  &  12.66203 &  869.2 &
146.8 & $7.2 \pm 0.6^b$ & SO\\
NGC 4395        & Sy1.5 & 0.0011 & 186.4539  &  33.54661 & 1103.5 & --
& $4.7 {+0.3 \atop -0.7}^i$ & V\\
3C 273          & QSO   & 0.1583 & 187.2779  &   2.05239 & 2004.6 &
299.2 & $9.81 {+0.1 \atop -0.07}^k$ & R\\
NGC 4507        & Sy2   & 0.0118 & 188.9023  &--39.90925 &  389.3 & --
& $7.6 \pm 0.6^c$ & SO\\
SWIFT J1238.9-2720 & Sy2 &0.0250 & 189.7271  &--27.30778 &   71.3 & --
& $8.6 \pm 0.5^n$ & KM\\
IGR J12391-1612 & Sy2   & 0.0367 & 189.7762  &--16.17975 &  618.7 &
39.4 & $8.9 \pm 0.5^n$ & KM\\
NGC 4593        & Sy1   & 0.0090 & 189.9143  & --5.34425 & 1466.5 &
213.5 & $6.99 {+0.08 \atop -0.1}^h$ & R\\
IGR J12415-5750 & Sy1.5 & 0.0242 & 190.3575  &--57.83417 & 1653.1 &
101.6 & $8.0 \pm 0.5^a$ & LL/CL\\
PKS 1241-399    & QSO   & 0.1910 & 191.1223  &--40.21289 &  551.3 & --\\
ESO 323-32      & Sy1   & 0.0160 & 193.3348  &--41.63717 &  744.4 &  23.9\\
3C 279          & BLLac & 0.5362 & 194.0465  & --5.78931 & 1144.1 &
185.0 & $8.4 \pm 0.5^e$ & CL\\
IGR J13000+2529$^x$& AGN& --     & 195.0000  &  25.48333 &  937.1 & 241.0\\
Mrk 783         & Sy1.5 & 0.0672 & 195.7452  &  16.40763 &  728.2 &  11.2\\
IGR J13038+5348 & Sy1   & 0.0302 & 195.9975  &  53.79172 &  366.3 &
16.5 & $7.5 \pm 0.5^n$ & KM\\
NGC 4945        & Sy2   & 0.0019 & 196.3587  &--49.47083 & 1100.0 &
141.8 & $6.2 \pm 0.3^l$ & M\\
IGR J13057+2036 & AGN   & --     & 196.4273  &  20.58103 &  853.1 &  12.0\\
ESO 323-77      & Sy1   & 0.0150 & 196.6108  &--40.41389 &  920.1 &
65.3 & $7.4 \pm 0.6^c$ & LL\\
IGR J13091+1137 & Sy2   & 0.025  & 197.2733  &  11.63414 &  391.1 &
8.2 & $8.6 \pm 0.5^n$ & KM\\
IGR J13109-5552 & Sy1   & 0.104  & 197.6795  &--55.86991 & 1506.6 &  26.6\\
NGC 5033        & Sy1.5 & 0.0029 & 198.3650  &  36.59358 &  743.2 &   6.2\\
IGR J13149+4422 & Sy2   & 0.0366 & 198.8155  &  44.40750 &  421.4 &  56.8\\
Cen A           & Sy2   & 0.0018 & 201.3651  &--43.01911 & 1268.0 &
149.5 & $8.0 \pm 0.6^c$ & K\\
ESO 383-18      & Sy2   & 0.0124 & 203.3596  &--34.01631 &  756.8 & 191.6\\
MCG-06-30-015   & Sy1.2   & 0.0077 & 203.9741  &--34.29558 &  767.2 &
213.0 & $6.7 {+0.1 \atop -0.2}^i$ & S\\
NGC 5252        & Sy2   & 0.0230 & 204.5667  &   4.54236 &   58.9 & --
& $9.03 {+0.4 \atop -0.02}^l$ & K\\
Mrk 268         & Sy2   & 0.0399 & 205.2964  &  30.37811 &  728.7 & --\\
4U 1344-60      & Sy1.5 & 0.0129 & 206.8833  &--60.61000 & 1537.7 & 231.2\\
IC 4329A        & Sy1   & 0.0161 & 207.3304  &--30.30956 &  381.9 &
52.6 & $\sim 7^g$ & R\\
Circinus Galaxy & Sy2   & 0.0014 & 213.2871  &--65.34084 & 2272.1 &
201.3 & $6.04 {+0.07 \atop - 0.09}^l$ & M\\
NGC 5506        & Sy1.9 & 0.0062 & 213.3120  & --3.20750 &  102.9 &
88.5 & $6.7 \pm 0.7^b$ & SO\\
IGR J14175-4641 & Sy2   & 0.076  & 214.2664  &--46.69419 & 1268.8 &  21.4\\
NGC 5548        & Sy1.5 & 0.0172 & 214.4985  &  25.13706 &  200.9 &
42.4 & $7.82 \pm 0.02^h$ & R\\
RHS 39          & Sy1   & 0.0222 & 214.8425  &--26.64472 &  454.4 &
8.1 & $8.7 \pm 0.5^n$ & KM\\
H 1426+428      & BLLac & 0.1291 & 217.1358  &  42.67472 &  499.5 &
224.7 & $9.1 \pm 0.7^e$ & SB\\
IGR J14471-6414 & Sy1   & 0.053  & 221.6158  &--64.27319 & 1797.1 & 171.8\\
IGR J14471-6319 & Sy2   & 0.038  & 221.8120  &--63.28868 & 1842.9 & 230.3\\
IGR J14492-5535 & AGN   & --     & 222.3038  &--55.60578 & 1886.0 & 142.4\\
IGR J14515-5542 & Sy2   & 0.018  & 222.8880  &--55.67733 & 1869.2 & 144.0\\
IGR J14552-5133 & NLS1  & 0.016  & 223.8223  &--51.57102 & 2007.3 &   7.9
& $6.3 \pm 0.5^p$ & LL \\
IGR J14561-3738 & Sy2   & 0.024  & 224.0342  &--37.64803 & 2185.9 & 340.8\\
\hline
IGR J14579-4308 & Sy2   & 0.016  & 224.4296  &--43.13000 & 2012.9 & 333.4\\
Mrk 841         & Sy1.5 & 0.0364 & 226.0050  &  10.43782 &  131.3 &
6.7 & $8.5 \pm 0.7^c$ & R\\
ESO 328-36      & Sy1   & 0.0237 & 228.6958  &--40.35861 & 2200.9 & 384.8\\
IGR J15161-3827 & Sy2   & 0.0365 & 229.0375  &--38.44806 & 2181.2 & 372.1\\
NGC 5995        & Sy2   & 0.0252 & 237.1040  &--13.75778 & 1129.2 &  28.6\\
IGR J15539-6142 & Sy2   & 0.015  & 238.3967  &--61.68206 & 1754.0 & 101.0\\
IGR J16024-6107 & Sy2   & 0.0114 & 240.4517  &--61.14822 & 1792.4 &  84.3\\
IGR J16056-6110 & Sy1.5 & 0.052  & 241.4643  &--61.19525 & 1727.2 &  64.4\\
IGR J16119-6036 & Sy1   & 0.016  & 242.9642  &--60.63194 & 1789.6 &  60.9\\
IGR J16185-5928 & NLS1  & 0.035  & 244.6518  &--59.45482 & 2066.0 &  44.6
& $7.4 \pm 0.5^p$ & LL \\
IGR J16351-5806 & Sy2   & 0.009  & 248.8071  &--58.08047 & 2230.3 & 39.3\\
IGR J16385-2057 & NLS1  & 0.0269 & 249.6250  &--20.94389 & 1505.9 &  80.4\\
IGR J16426+6536 & NLS1  & 0.323  & 250.7670  &  65.54747 &   56.0 &
--  & $7.0 \pm 0.5^c$ & LL/CL\\
IGR J16482-3036 & Sy1   & 0.0313 & 252.0623  &--30.58502 & 2450.9 &  53.7\\
ESO 138-1       & Sy2   & 0.0091 & 252.8333  &--59.23389 & 1509.2 &  21.4\\
NGC 6221        & Sy2   & 0.0050 & 253.1942  &--59.21639 & 1481.0 &  21.4\\
NGC 6240        & Sy2   & 0.0245 & 253.2457  &   2.40047 &  300.5 &  67.7\\
Mrk 501         & BLLac & 0.0337 & 253.4676  &  39.76017 &  476.3 &
70.6 & $9.2 \pm 0.3^e$ & S\\
IGR J16558-5203 & Sy1   & 0.054  & 254.0234  &--52.06135 & 3065.7 & 249.6
& $7.9 \pm 0.5^p$ & LL \\
IGR J16562-3301 & BLLac & --     & 254.0701  &--33.03680 & 3669.2 & 111.4\\
NGC 6300        & Sy2   & 0.0037 & 259.2467  &--62.81972 &  237.9 &
8.9 & $5.5 \pm 0.4^c$ & X\\
IGR J17204-3554 & AGN   & --     & 260.1042  &--35.90000 & 6318.5 & 292.3\\
QSO B1730-130$^x$& QSO  & 0.9020 & 263.2613  &--13.08042 & 1080.1 &  33.1\\
GRS 1734-292    & Sy1   & 0.0214 & 264.3681  &--29.13403 & 9122.1 & 665.7
& $8.9 \pm 0.7^q$ & SO \\
IGR J17418-1212 & Sy1   & 0.0372 & 265.4625  &--12.19611 & 1300.1 &  45.5\\
IGR J17488-3253 & Sy1   & 0.020  & 267.2297  &--32.91449 & 6159.4 & 410.4\\
IGR J17513-2011 & Sy2   & 0.047  & 267.8068  &--20.20405 & 6220.1 & 268.7
& $6.0 \pm 0.5^p$ & LL \\
IGR J18027-1455 & Sy1   & 0.0034 & 270.6974  &--14.91522 & 2463.8 & 173.0 \\
IGR J18244-5622 & Sy2   & 0.0169 & 276.0812 & --56.36909 &  103.3 &   4.4\\
IGR J18249-3243 & Sy1   & 0.355  & 276.2361 & --32.71661 & 5126.8 & 208.7\\
IGR J18259-0706 & Sy1?  & --     & 276.48958&  --7.17264 & 2009.5 & 226.1\\
PKS 1830-211    & BLLac & 2.5070 & 278.4162 & --21.06106 & 2775.4 & 111.2\\
3C 382          & Sy1   & 0.0579 & 278.7641 &   32.69635 &   15.2 &
0.7 & $9.2 \pm 0.5^n$ & KM\\
ESO 103-35      & Sy2   & 0.0133 & 279.5846 & --65.42805 &   28.5 & --
& $7.1 \pm 0.6^c$ & X\\
3C 390.3        & Sy1   & 0.0561 & 280.5374 &   79.77142 &  448.0 &
33.2 & $8.46 {+0.09 \atop -0.1}^g$ & R\\
ESO 140-43$^*$  & Sy1   & 0.0141 & 281.2917 & --62.35583 &  378.0 & 27.7\\
IGR J18559+1535 & Sy1   & 0.084  & 284.0000 &   15.63694 & 2594.9 &  24.7\\
ESO 141-55$^*$  & Sy1   & 0.0366 & 290.3092 & --58.67083 &   378.0
& 79.9 & $7.1 \pm 0.6^c$ & SO\\
1RXS J192450.8-29143& BLLac & 0.3520 & 291.2127 &--29.24170&955.7 &  28.6\\
1H 1934-063     & Sy1   & 0.0106 & 294.3879 &  --6.21806 &  654.2 &
2.1 & $7.9 \pm 0.6^c$ & SO\\
IGR J19405-3016 & Sy1   & 0.052  & 295.0631 & --30.26347 &  930.3 &  22.3\\
NGC 6814        & Sy1.5 & 0.0052 & 295.6683 & --10.32333 &  438.5 &
7.4 & $7.1 \pm 0.2^h$ & CL\\
IGR J19473+4452 & Sy2   & 0.0539 & 296.8307 &   44.82845 & 1194.3 &   1.0\\
3C 403          & Sy2   & 0.0590 & 298.0617 &    2.50778 &  490.2 & --\\
QSO B1957+405   & Sy2   & 0.0561 & 299.8682 &   40.73386 & 2258.1 &
69.5 & $9.4 \pm 0.1^l$ & K\\
1ES 1959+650$^x$    & BLLac & 0.048  & 299.9994 &   65.14851 &   11.2 & --
& $8.1 \pm 0.3^e$ & S\\
ESO 399-20      & NLS1  & 0.0250 & 301.7383 & --34.54833 &  857.2 &  24.7\\
IGR J20187+4041 & Sy2   & 0.0144$^r$& 304.6606 &   40.68344 & 2851.0 & 442.7\\
IGR J20286+2544 & Sy2   & 0.013  & 307.1462 &   25.73361 &  840.2 &   4.1\\
4C 74.26$^x$    & QSO   & 0.1040 & 310.6549 &   75.13403 &   77.1 &
-- & $9.6 \pm 0.5^e$ & CL\\
Mrk 509         & Sy1.2 & 0.0344 & 311.0406 & --10.72348 &   71.7 &
57.3 & $8.16 {+0.03 \atop -0.04}^g$ & R\\
S5 2116+81      & Sy1   & 0.086  & 318.5021 &   82.07975 &  176.5 &
-- & $8.8 \pm 0.5^n$ & KM\\
IGR J21178+5139 & AGN   & --     & 319.4468 &   51.64823 & 1261.9 & 171.3\\
IGR J21247+5058 & Sy1   & 0.020  & 321.1640 &   50.97329 & 1392.7 & 174.7\\
IGR J21272+4241$^x$ & Sy1.5 & 0.316  & 321.7917 &   42.69194 &  836.3 & 189.8\\
IGR J21277+5656 & Sy1   & 0.0144 & 321.9373 &   56.94436 &  984.5 & 166.6\\
RX J2135.9+4728 & Sy1   & 0.0252 & 323.9766 &   47.47453 & 1213.4 & 136.3\\
PKS 2149-306    & FSRQ  &  2.345 & 327.9813 & --30.46492 &  258.6 &  47.1\\ 
NGC 7172        & Sy2   & 0.0086 & 330.5071 & --31.87167 &  346.5 &
68.8 & $7.7 \pm 0.6^c$ & SO\\
BL Lac          & BLLac & 0.0686 & 330.6804 &   42.27778 &  738.1 &
-- & $8.2 \pm 0.7^e$ & SB\\
IGR J22292+6647 & Sy1$^o$   & 0.113$^o$ & 337.3062 &   66.78106 & 1678.9 &   9.3\\
NGC 7314        & Sy1   & 0.0048 & 338.9419 & --26.05047 &  546.8 &
1.7 & $6.0 \pm 0.5^c$ & S\\
Mrk 915         & Sy1   & 0.0241 & 339.1938 & --12.54517 &  609.0 &  12.8\\
IGR J22517+2217 & BLLac & 3.668  & 342.9280 &   22.29900 &  251.3 &   5.8\\
3C 454.3        & BLLac & 0.8590 & 343.4906 &   16.14822 &  207.1 &
36.3 & $9.2 \pm 0.7^e$ & CL\\
\hline
1H 2251-179     & Sy1   & 0.0640 & 343.5245 & --17.58203 &  579.2 &
314.6 & $<6.9^c$ & W\\
NGC 7469        & Sy1   & 0.0163 & 345.8156 &    8.87386 &  139.2 &
-- & $7.09 \pm 0.05^g$ & R\\
MCG-02-58-022   & Sy1.5 & 0.0469 & 346.1812 &  --8.68572 &  624.5 &
28.2 & $7.1 \pm 0.6^c$ & SO\\
NGC 7603$^x$    & Sy1.5 & 0.0295 & 349.7359 &    0.24347 &  104.5 &
-- & $8.1 \pm 0.3^e$ & S\\
IGR J23206+6431 & Sy1   & 0.0732 & 350.15   &   64.52    & 3888.2 & 349.8\\
IGR J23308+7120 & Sy2   & 0.037  & 352.6552 &   71.37911 & 1965.1 &  --\\
IGR J23524+5842 & Sy2   & 0.164  & 358.0917 &   58.75908 & 4076.7 & 361.1\\
\hline
\end{tabular}
\medskip
$^{*}$ Beckmann et~al. (\cite{Beckmann07b}); $^a$ Masetti et~al. (\cite{Masetti09}); $^b$ Bian \& Gu (\cite{Bian07}); $^c$ Middleton et~al. (\cite{Middleton08});
$^d$ Kaspi et~al. (\cite{Kaspi00}); $^e$ Woo \& Urry (\cite{Woo02}); $^f$ Merloni et~al. (\cite{Merloni03}); $^g$ Peterson et~al. (\cite{Peterson04}); $^h$ Hicks \& Malkan (\cite{Hicks08});
$^i$ Uttley \& McHardy (\cite{Uttley05}); $^k$ Paltani \& T\"urler (\cite{Paltani05}); $^l$
Graham (\cite{Graham08}); $^m$ \mbox{Zurita-Heras} et~al. (\cite{Zurita09}); $^n$ Winter et~al. (\cite{Winter09});
$^o$ Butler et~al. (\cite{Butler09}); $^p$ Masetti et~al. (\cite{Masetti06}); $^q$ see Appendix~\ref{section:singlesource}; $^r$ Goncalves et~al. (\cite{Goncalves09}); $^x$ not detected in the data set presented here.
\end{table}